Author's response to reviews

Title: Utilization of Base Deficit and Reliability of Base Deficit as a Surrogate for Serum Lactate in the Peri-Operative Setting

Authors:

Lakhmir S Chawla (lchawla@mfa.gwu.edu)
Amirali Nader (amirali_nader@yahoo.com)
Todd Nelson (tanelson@gmail.com)
Trusha Govindji (tgovin@gwu.edu)
Ryan Wilson (gwryanwilson@yahoo.com)
Sonia Slyzk (soniaszlyk@hotmail.com)
Aline Nguyen (aline312@hotmail.com)
Christopher Junker (cjunker@gmail.com)
Michael Seneff (mseneff@mfa.gwu.edu)

Version: 2 Date: 1 December 2009

Author's response to reviews:

We have made the revisions suggested.